



Article Policy Implications of the Clean Heating Transition: A Case Study of Shanxi

Eun Ju Lee¹, Moon Joon Kim² and Jae-Seung Lee^{3,*}

- ¹ Graduate School of Energy and Environment (KU-KIST Green School), Korea University, Seoul 02481, Korea; eunju_lee@korea.ac.kr
- ² Energy Environment Policy and Technology, Graduate School of Energy and Environment (KU-KIST Green School), Korea University, Seoul 02481, Korea; moonjoonkim@korea.ac.kr
- ³ Graduate School of International Studies, Korea University, Seoul 02841, Korea
- * Correspondence: jaselee@korea.ac.kr

Abstract: This study provides empirical evidence of the impact of recentralized governance on environmental performance by examining the implementation of a clean heating transition. It investigated the impact of a centralized clean heating transition on sulfur dioxide (SO_2) levels in Shanxi province from January 2015 to March 2021. Using a difference-in-differences approach, this study found that the centralized clean heating transition led to a significant improvement in air quality; however, the excessive response of Shanxi province prevented adequate heating supply for residents. As a result, the Chinese government had to reverse its initial plans for a coal ban and the promotion of gas plants. This outcome implies that recentralization cannot control the autonomy of local governments in responding to and achieving the central targets, even though it may provide incentives to prioritize environmental issues. The recentralization proved to be ineffective, in contrast to what was theoretically anticipated, and even undermined the energy transition efforts.



Citation: Lee, E.J.; Kim, M.J.; Lee, J.-S. Policy Implications of the Clean Heating Transition: A Case Study of Shanxi. *Energies* **2021**, *14*, 8431. https://doi.org/10.3390/en14248431

Academic Editor: David Borge-Diez

Received: 20 October 2021 Accepted: 22 November 2021 Published: 14 December 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). **Keywords:** APACCP; clean heating transition; SO₂; Shanxi province; difference-in-differences; cadre responsibility and evaluation system; target-based implementation; recentralization

1. Introduction

Coal is the primary energy source for winter heating in China; however, it is the main contributor to winter haze [1,2]. The heating system in China was established during the planned economy in the 1950s. Because it was considered public welfare, the central government provided free or highly subsidized coal for heating [3]. Due to limited financial and energy resources, the coal-fired centralized heating system was constructed only in 14 provinces and partially in Henan province, which were north of the border of the Hui River and Qin Mountains [4]. Therefore, cities and provinces such as Shanghai, Jiangsu, and Anhui, which are south of this border, were excluded from the heating zone, even though their average winter temperatures are between 0-10 °C.

Unfortunately, this leads to Chinese citizens experiencing difficult winters. Heating system can be categorized into centralized and decentralized. Centralized heating is supplied by the district and is obtained from coal-fired boilers and combined heat and power plants. Decentralized heating includes heating using individual boilers and coal furnaces, known as the Chinese Kang (stove). The output from most of the district-supplied heating systems are dispatched to urban areas, whereas the majority of rural houses use individual heating. As per official records, approximately seven to eight percent of the total coal consumption in China has been recorded in winter for heating; however, the scattered coal consumption for individual heating has not been fully calculated using official statistics [5,6]. In 2016, the Chinese government publicized the amount of coal consumption for heating as 400 million tons of coal equivalent (Mtce), with the scattered coal consumption accounting for half that amount [4].

Both centralized and decentralized systems have been criticized for predominantly using coal and as being the main contributors of winter air pollution. Firstly, the majority of centralized heating bills continue to be calculated based on the heated area instead of consumption. Hence, even though centralized heating is only available during certain winter periods, the residents are less incentivized to control their usage because they are charged a fixed amount that is independent of the consumption. Secondly, a centralized heating system does not allow individual users to control the temperature; users open windows to adjust room temperature if their rooms are too hot [7]. In contrast, people who reside in houses that lack a temperature control device must endure uncomfortable low temperatures during winters. According to a study, such poor control of room temperature wastes 20–30% of the energy used [8]. Thirdly, Chinese Kangs (stoves), the primary heating equipment for rural residents, are not equipped with air pollution abatement devices and consequently, emit more air pollutants. Fourthly, poor-quality scattered coal is the main fuel used for decentralized heating in rural areas. In addition to having a low combustion efficiency, it also contains higher amounts of ash, sulfur and volatile matter [6]. Therefore, the SO₂ emission factors of scattered coal used in residential stoves and industrial boilers are approximately 11 and 8 times higher, respectively, compared to those of the coal used in centralized power generators [9].

Since early 2013, the capital city of Beijing has suffered from haze episodes with high concentrations of particulate matter, especially in winter. Given the significant adverse influence of winter haze on public health and social stability, the Chinese government has increasingly emphasized air pollution control. The State Council published the Action Plan for Air Pollution Prevention and Control (2013–2017) (hereafter APAPPC 2013–2017) and outlined ten measures (such as enhancing overall treatment and reducing discharges of multiple pollutants, adjusting and optimizing industrial and energy structures, and increasing the supply of clean energy) to improve the nation-wide air quality. The Chinese premier, Li Keqiang, declared 'war on smog' in his first government work report at the 2014 National People's Congress and prioritized the air pollution agenda; however, air pollution episodes have frequently been witnessed [10,11]. The Beijing municipal government was forced to issue the first-ever 'red alert' in December 2015 and again in December 2016. According to the Heavy Air Pollution Contingency Plan of Beijing, the red alert should be announced when authorities forecast more than three consecutive days of severe air pollution associated with air quality index (AQI) above 300 [12]. Following the red alert-related emission reduction measures, schools and kindergartens were suspended, enterprises and institutions were asked to implement flexible work systems, only half the vehicles were allowed to run under the odd-even license plate policy, all outdoor construction work was put off, and listed factories that contributed to pollution were shut down temporarily [12,13].

The central government publicized the urgency and seriousness of the issue, as the air pollution in Beijing was severe until a year before the deadline of the APACCP 2013–2017. At the 14th meeting of the Central Financial and Economic Leading Group in December 2016, Chinese President Xi Jinping stressed on using clean energy for winter heating in northern China to ensure a warm winter for the people and reduce heavy pollution days [14]. The Leading Group is the core of the Communist Party of China (CPC) that governs the economy and discusses related issues. This unprecedented and unexpected statement addressing environmental issues clearly implied that the elite leadership recognized heavy air pollution as detrimental to economic growth and as an urgent problem to solve. Consequently, a series of enhanced policies and measures were issued in 2017 to fulfill the targets set for the APACCP 2013–2017, primarily promoting the coal-to-gas, coal-to-electricity transition. Both President Xi and Premier Li reiterated the importance of environmental protection and clean heating transition in 2017. Moreover, as the initial transition targets continuously increased and environmental inspections were strengthened by the central leadership, the transition was highly centralized.

Because the Chinese government has assigned performance targets to environmental and energy concerns under the cadre responsibility and evaluation system to enhance policy implementation, questions on the effectiveness of target-based implementation have been centered on studies of the environmental governance in China. Theoretically, the cadre responsibility and evaluation system would incentivize local cadres and officials to prioritize environmental policies because environmental performance had been added to the cadre evaluation criteria. However, existing literature suggests that the targetbased approach creates an implementation gap due to poor incentives and structural shortcomings. Because the Chinese central government places greater importance on economic growth than environmental protection, local governments do not have sufficient incentives to implement its environmental policies and prioritize economic tasks that materialize their performance in the short term [15]. The cadre has a high turnover rate due to the central government repressing local protectionism. Hence the officers have limited time to impress higher-level governments and advance their careers [16]. Central state-owned enterprises (SOEs), which have long defied environmental regulations under the protection of the central government in order to generate better economic outcomes, have tacitly supported local protectionism. Environmental protection policies have loosely enforced local enterprises in pursuit of better economic outcomes for promotion [17]. In summary, local cadres and officials have put more effort into what their superiors prioritize [18].

Another group of existing literature points out the structural shortcomings that arise due to the local governments being held responsible for the tasks of implementation, monitoring, and verification of environmental policies. Therefore, the cadre responsibility and evaluation system itself have caused many endogenous problems, such as lack of reliable energy data, data falsification, and collusion to fake the environmental performance [19–21]. Two scholars explicitly analyzed these structural shortcomings and described it as "command without control"; the central government sets the policies and performance targets but does not control its implementation and monitoring [22]. Owing to "blaming politics," the central government shifts the responsibilities of implementation and monitoring onto local governments to avoid blame should the environmental policies fail [23]. These studies highlight that the central government must have a clearer intention of environmental protection and an enhanced inspection system to narrow the gap and successfully achieve the intended goals.

As was mentioned earlier, the importance of the clean heating transition on environmental protection and centralized environmental inspection has been supported by the highest authority. However, the impact of the recentralized Chinese environmental governance on air quality and its environmental outcomes have not been sufficiently analyzed; only a few studies have empirically and theoretically studied its impacts on environmental performance [24]. This study seeks to analyze the empirical evidence on the impact of recentralized governance on environmental performance by examining the implementation of the clean heating transition. It was observed that the centralized clean heating transition tasks led to a significant improvement in air quality since the policy intervention, but adequate heating was prevented due to the rapid and large-scale transition. This implies that recentralization may have been able to provide proper incentives to repress local protectionism, but it could not control the autonomy of local governments in responding to and achieving the central targets. As a result, recentralization was not as effective as theoretically anticipated.

The remainder of this paper is structured as follows: Section 2 provides policy background to support the study design. Section 3 briefly reviews the role of cadre responsibility and the evaluation system in Chinese environmental governance and its impact on the policy outcomes. Section 4 explains why this study chose Shanxi province and illustrates the study design and methodology. Section 5 presents the empirical results of the research questions. Section 6 discusses the policy output induced desirable and undesirable outcomes, namely clean but cold winters. A summary of the key findings and policy implications has been provided in Section 7.

2. Policy Background

In September 2013, the APACCP 2013–2017 was issued with ten measures to improve the nation-wide overall air quality. The goal of the APACCP 2013–2017 was to reduce the PM10 concentrations to 10% of the 2012-levels by 2017 in all the cities with prefectureand above-levels. The three main regions of Beijing-Tianjin-Hebei, Yangtze delta, and Pearl River delta received the higher PM2.5 reduction targets of 25%, 20%, and 15%, respectively [25]. The peak PM2.5 concentrations might have been reduced by the targets; however, high PM2.5 concentrations were observed during the heating seasons. In fact, the coal switch in the residential sector was mentioned but not strongly encouraged in the APACCP 2013–2017. Accordingly, the Measures to Evaluate the Implementation of the APACCP 2013–2017 only set the evaluation score for the clean management of scattered coal in Beijing, Tianjin, and Hebei provinces. The rest of the northern heating zone was exempted from being assessed for any clean heating-relevant missions [26]. Frequent haze became a source of public unrest as ordinary lives were affected by actions such as closing schools and factories, shutting down highways and airports, and the Chinese government had to take strong action against winter haze.

A series of measures and plans were introduced during the ending year of the APACCP 2013–2017 to encourage meeting the performance targets. Promotion of the clean heating transition was placed in the center of the policy documents. Firstly, the Work Plan for Air Pollution Prevention and Control for the Beijing-Tianjin-Hebei Region and Surrounding Areas in 2017 (hereafter WPACCP 2017), was issued in February 2017. The Ministry of Environmental Protection (MEP) (superseded by the Ministry of Ecology and Environment (MEE) in March 2018), based on extensive fieldwork and model simulation, selected Beijing, Tianjin, and its surrounding 26 cities, and formed a regional air pollution transmission channel that significantly influenced the PM2.5 concentrations of Beijing. The 26 neighboring cities surrounding Beijing and Tianjin are Shijiazhuang, Tangshan, Langfang, Baoding, Cangzhou, Hengshui, Xingtai, and Handan in Hebei province; Taiyuan, Yangquan, Changzhi, and Jincheng in Shanxi province, Jinan, Zibo, Jining, Dezhou, Liaocheng, Binzhou, and Heze in Shandong province; and Zhengzhou, Kaifeng, Anyang, Hebi, Xinxiang, Jiaozuo, and Puyang in Henan province. Since the formation of the channel, unified emission reduction measures were planned to be carried out in these cities simultaneously to reduce extremely high PM2.5 concentrations in Beijing during heavy pollution episodes. Each city was tasked with completely replacing the coal heating system by 0.05 million to 0.1 million households until October 2017 [27]. The total number ranged from a minimum of 1.4 million to a maximum of 2.8 million households.

The urgency and significance of the clean heating transition were heightened in the second half of 2017, as more rigid measures were introduced. In August 2017, higher targets for the clean heating transition were set for the 2 + 26 key cities throughout the Action Plan for Comprehensive Management of Air Pollution in the Beijing-Tianjin-Hebei and Surrounding Areas in the Autumn and Winter of 2017–2018 (hereafter APCMAP 2017–2018). The Action Plan upgraded the targets with detailed measures aimed at completely replacing coal with gas or electricity in more than 3.55 million households by October 2017 [28]. In December 2017, the National Development and Reform Commission (NDRC) and nine other ministries and commissions jointly issued the Winter Clean Heating Plan for Northern China 2017–2021 (hereafter CWHP 2017–2021). As can be inferred from the name of the plan, it covers all of northern China. It set targets to complete the transition for more than 26 million households and save 150 million Mtce of scattered coal by 2021 in the residential sector. The CWHP 2017–2021 specifically emphasized that the 2 + 26 key cities should lead the transition and allocate higher targets of transition rate compared to other northern heating areas. The clean heating rate in northern China must reach 50% by 2019 and 70% by 2021, at least 90% in the urban areas of the 2 + 26 key cities, above 70% in the prefecture

and urban-rural fringe areas, and more than 40% in the rural by 2019. The targets were 100%, 80%, and 60% respectively for the aforementioned areas by 2021 [4].

The State Council issued the Three-Year Action Plan for Winning the Blue Sky Defense War 2018–2020 (hereafter TYAP 2018–2020) as the APACCP 2013–2017 follow-up policy in June 2018. The three key battle fields were adjusted to the Beijing-Tianjin-Hebei and surrounding areas (Shandong, Shanxi, and Henan provinces), the Yangtze delta, and the Fen-Wei Plain, which stretches over the Shanxi, Shaanxi, and Henan provinces [29]. As the Fen-Wei Plain region is located in the northern heating zone, the 11 cities belonging to the region also started to take serious action for switching to coal. The efforts of the Chinese government to prevent and control winter haze have continued. Since the APCMAP 2017–2018 was published, the yearly transition targets were introduced to support the transition through APCMAP in the Beijing-Tianjin-Hebei and Surrounding Areas in the Autumn and Winter of 2018–2019 and 2019–2020. These two documents were combined in 2020; in total, 39 focal cities are intensively working on the transition tasks through APCMAP in the Beijing-Tianjin-Hebei and Surrounding Areas and the Fen-Wei Plain Region in the Autumn and Winter of 2018–2019 and 2019–2020. These two documents were combined in 2020; in total, 39 focal cities are intensively working on the transition tasks through APCMAP in the Beijing-Tianjin-Hebei and Surrounding Areas and the Fen-Wei Plain Region in the Autumn and Winter of 2020–2021.

3. Cadre Responsibility and Evaluation System in Chinese Environmental Governance

3.1. The Role of Cadre Responsibility and Evaluation System in Chinese Environmental Policy Implementation

The Chinese central government has been strengthening and maintaining its managerial control over policy implementation through the cadre responsibility and evaluation system, which is built upon the nomenklatura system [15,30,31]. The leading positions in the Party and government listed in the nomenklatura are selected and appointed by the Communist Party of China (CPC). The Party committees also exercise authority over the promotion, removal and transfer of leading cadres. The decisions on promotion, removal, and transfer are made using the cadre responsibility and evaluation system, which evaluates the work performance of the leading cadres [31,32]. Therefore, the system itself distributes political incentives to motivate local leaders who are responsible for implementing policies and measures to carry out the central government's policy priorities.

To assess the work performance of local leaders, performance contracts and targets are used under the cadre responsibility and evaluation system. The performance contracts, known as target responsibility letters, are signed between higher- and lower-level governments (i.e., the central and provincial governments, provincial governments and prefecture-level cities, and prefecture-level cities and counties, counties, and townships). Through the letters, the leaders of lower-level governments pledge to fulfill the targets assigned by higher levels and to take personal responsibility to achieve them [31]. The central government assigns numerical national performance targets, allocates the targets to provinces, and creates a scoring system to weigh the performance. These performance targets are ranked in the priorities of the central government as soft-expected (yuqixing) and hard-binding (yueshuxing) targets in the national five-year plans. The binding hard targets with veto power (yipiao foujue) have the highest priorities; if the leader fails to attain these targets, all their other achievements will lose validity [15,16,19,31,33].

The Chinese environmental policy implementation relied on cadre responsibility and an evaluation system to encourage target accomplishment. The 11th Five-Year Plan for National Economic and Social Development (2006–2010) set the first national environmental binding targets. Environmental targets were included in the previous five-year plans, but they were largely ignored, as evidenced by the SO₂ emissions increasing by 27% during the 10th Five-Year Plan (2001–2005) periods [19]. The 10% reduction target was completely ignored. The Chinese government upgraded the environmental targets from expected to binding status. It also distributed the national SO₂ emission reduction targets of the 11th Five-Year Plan to provincial governments through the National Control Plan for the Total Emission of Major Pollutants during the 11th Five-Year Plan Period in 2006. In addition, the State Council delivered the Method for Assessing the Total Emissions Reduction of Major Pollutants to announce the "one-vote veto (one strikeout)" system in 2007 [34,35]. Consequently, compliance with the assigned binding targets became the key to performance assessment and evaluation of local cadres. The failure to meet these "veto targets" could negate all positive credits acquired from other cadre performances. Therefore, local cadres who performed well received rewards such as promotion and financial support for future projects [33], but local cadres who failed to meet the targets were penalized by demotion and suspending the review and application for new projects [30].

In addition to SO₂, the reduction of PM2.5 had begun to be controlled by the cadre responsibility and evaluation system since 2013. The 12th Five-Year Plan (2011–2015) had presented more detailed targets on environmental and climate change issues. The emission target for NOx, another major air pollutant, has been set, but the PM2.5 concentration reduction targets were omitted. The Chinese government published the APAPPC 2013–2017 to control PM2.5. Provincial Party secretaries and governors of all municipalities and provinces with targets in this document were asked to sign letters of target responsibility for tackling air pollution. Then, the Measures to Evaluate the Implementation of Action Plan on Air Pollution Prevention and Control 2013-2017 was issued, and a scoring system was introduced to evaluate the performance of local officials in the key cities of work listed in the APACCP 2013–2017 [36]. The Measures specified that achieving reduction is a mandatory indicator in the performance assessment of the three key regions mentioned in the APACCP 2013–2017, and Shanxi, Shandong, and Inner Mongolia were also asked to control PM2.5. Later that year, PM2.5 concentration reduction targets were finally added in the 13th Five-Year Plan in 2016. The Chinese government has continuously introduced stronger enforcement measures to ensure implementation and incentivize local cadres to follow the policy preferences of the central government.

3.2. The Cadre Responsibility and Evaluation System and Its Impact on Policy Outcomes

Because environmental policy implementation has been linked to cadre responsibility and evaluation system, environmental issues began to be placed at the top of the policy agenda, with increased responses from the officials. However, questions on its effectiveness have been raised constantly, as the implementation gap remains a key issue in China's environmental governance. Here, the implementation gap can be understood as the difference between the long-term environmental goals of the central government and implementation outcomes at the local level [15,18,37]. In other words, under the Chinese environmental governance, local governments achieve performance targets, but these policy outputs frequently do not extend to positive policy outcomes. The poor incentives and structural shortcomings mentioned in the earlier section have distorted the implementation process for the following reasons.

Firstly, local Chinese cadres tend to adopt a last-minute approach for environmental policy implementation. The central government sets the rules and allocates targets; however, whether its interests or intentions are to improve environmental outcomes is doubtful [15]. More local leaders were promoted because of better economic performance and not environmental protection [38]. In addition, they often face many constraints, such as financial pressure, lack of technology, and limited political capacity to enforce environmental policies. There is significant financial pressure; for example, the installation and operation of pollution removal or reduction equipment require additional investment, but benefits mostly occur in the longer term [16]. Consequently, local governments have not been sufficiently incentivized to prioritize environmental policies to impress their superiors, as their terms in office do not last as long as five years. However, because environmental binding targets must be achieved, efforts to meet the targets are intensified or rushed when the deadline approaches. Because the targets need to be achieved in a limited time, the last-minute response misleads the implementers to manipulate data or take short-sighted action [39]. The damage to public interests at the end year of the 11th Five-Year Plan was due to the misconduct of many localities to achieve environmental and energy targets. Finally, an emergency note prohibiting short-term electricity cuts and closure of production lines was issued by the NDRC in September 2010 [39]. Unfortunately, the pollution became severely visible from the beginning year of the 12th Five-year Plan in 2011, and many local governments sought to attract outside enterprises to increase GDP so that they could easily improve the energy intensity ratio.

Secondly, the local government cadres tend to play a "numbers game" as the environmental improvements have been monitored and evaluated at the local level. For cadre performance evaluation, environmental data must be accurately and completely collected, reported, and verified. Since the central government decentralized this monitoring process, it has been accused of providing substantial room to local cadres formulating "satisfying data" [19,20,24]. Elevating environmental targets to binding status also leads local officials to be more creative in fabricating and falsifying data, as they do not have incentives to report failures of implementation, which could harm their career and chances for advancement [22]. The local environmental protection bureaus (EPBs), which are responsible for pollution control, environmental monitoring, and ecological conservation, are considered to have the highest authority and responsibility over environmental policy implementation [15]. However, in practice, the local EPBs are closely linked to local governments as the bureaus are run using the local fiscal budget, and the EPB officials are appointed by local leaders. Local fiscal incomes have been mainly raised by the sales and value-added taxes of local enterprises since the 1994 tax reform. Local governments protect them from generating higher tax revenues while not being disturbed by environmental regulations. Therefore, local environmental officials are strongly influenced by and unavoidably protect their local protectionist leaders from being poorly assessed [40]. This trend also extends to the dynamic between central inspectors and local governments. Because the MEP could face significant consequences if the national environmental targets are not met, the central inspectors overlook debatable or non-existent reductions reported by local governments when the national targets seem difficult to achieve [19]. Since 2016, environmental inspections have been centralized. Its influence and effectiveness in combating air pollution will be discussed in a later section.

Finally, Chinese local government officials are not incentivized to surpass the binding targets, and neglect untargeted environmental issues. Failing to meet binding targets could prevent local officials from being promoted or evaluated as excellent; hence, they have strong incentives to meet the targets. However, they are not motivated to surpass targets, as regulations do not clearly mention or offer different reward structures for provinces that fulfill and exceed targets. Therefore, binding environmental targets tend to provide the baseline requirements. For example, the Method for Assessing the Total Emissions of Major Pollutants issued in 2007 directed that provinces that achieved targets would be rewarded and prioritized to receive financial support from the central government for subsequent environmental tasks [34,41]. Similarly, the Measures to Evaluate the Implementation of Action Plan on Air Pollution Prevention and Control 2013–2017 stated that the assessment results would influence financial support for provinces, differentiating between provinces with excellent and incomplete statuses [26]. Again, rewards for provinces that exceeded the targets were not mentioned. Moreover, during interviews with local officials, Genia Kostka revealed that they tend to purposefully fulfill only the minimum requirements to leave room for future targets or not to get higher targets for the next term [22,39]. As the target allocation method is undisclosed, they were worried that performance history could be a possible standard for the next target allocation. Moreover, since the number of environmental and energy obligatory tasks increased from five in the 11th Five-Year Plan to 11 and 16 in the 12th and 13th Five-Year Plans, respectively, untargeted environmental issues and concerns were easily ignored as they would only add "additional work" to local cadres. Hence, even though local leaders had recognized the seriousness of the problem due to particulate matter, they were not seriously considered until the binding targets of PM10 and PM2.5 were listed in the policy plans [22].

4. Methodology and Data

4.1. Case Study: Shanxi Province

Shanxi is a major coal-producing province in China. It was ranked as the best coalproducing province until 2015, when Inner Mongolia achieved the highest rank; however, Shanxi recaptured its former ranking in 2020 [5]. The contribution of Shanxi's coal industry to local GDP was the highest at 32.02% in 2016. Inner Mongolia was placed second, with a contribution of only 18.37% [42]. Shanxi has inevitably been using more coal as it is abundant; coal consumption accounted for 84.6% of the primary energy mix in 2017 [43]. Therefore, compared to the other six municipalities and provinces of the 2 + 26 key cities and the Fen-Wei Plain region, the per capita coal consumption of Shanxi was the highest (Figure 1), which has to be reduced for air pollution control; hence, the province should be encouraged to make the low-carbon transition.



Figure 1. Coal consumption per capita in 7 municipalities and provinces. Source: China Statistical Yearbook 2014–2020, China Energy Statistical Yearbook 2014–2020.

The 2 + 26 key cities and the 11 cities of the Fen-Wei Plain region are the most polluted regions to which the central government has paid more attention. The former received its transition targets in 2017 and the latter in 2018. Therefore, these regions were inspected more intensively than the untargeted regions to ensure their environmental performance. Among the seven municipalities and provinces of the 2 + 26 key cities and the Fen-Wei Plain region, only Shanxi and Henan provinces belong to both groups. However, as was mentioned earlier, Henan province is partly placed in the northern heating zone; hence, Shanxi province is more suitable to examine the impact of the clean heating transition on SO_2 levels.

This study classified the 11 prefecture-level cities of Shanxi into three groups. The four 2 + 26 key cities, Taiyuan, Yangquan, Changzhi, and Jincheng, and four cities of the Fen-Wei Plain region, Jinzhong, Yuncheng, Linfen, and Lüliang, were placed in treatment groups 1 and 2, respectively. The remaining three cities, Datong, Xinzhou, and Shuozhou, which had not received any performance targets from the central government, were placed in the control group (Figure 2).



Figure 2. Shanxi province and its three groups.

4.2. Study Design and Data Sources

This study used daily concentrations of SO₂ as the primary assessment index to examine changes in air quality due to policy interventions. Particulate matter is formed when primary pollutants, such as SO₂ and other pollutants, react in the atmosphere. Firstly, SO₂ needs to be controlled in China to reduce PM10 and PM2.5. However, the driving factors of PM2.5 concentrations are not solely limited to coal combustion. Therefore, the SO₂ level is a more appropriate indicator to examine the impact of performance targets on the clean heating transition. SO₂ emission reduction was one of the first national environmental tasks with a binding target set in the 11th Five-Year Plan (2006–2010), and the task continued to the 13th Five-Year Plan (2016–2020). The APCMAP 2017–2018 and TYAP 2018–2020 of Shanxi autonomously added SO₂ concentration reduction targets instead of emission reduction targets in the Five-Year Plans [44,45]. The APCMAP 2017–2018 of Shanxi set a target for reducing the SO₂ concentration by 40% compared to the same period of the previous year, and TYAP 2018–2020 assigned a 50% reduction target compared to the levels in 2015.

 SO_2 is a key gaseous tracer for coal consumption. Because heating is the most significant source of additional air pollution in winter and coal is the main energy source for heating, approximately 83% in northern China, considerable SO_2 reductions would be achieved through the clean heating transition [46]. Meng et. al proposed that if "there were no significant changes in production and emissions of industry, power generation and other sectors, the extra emissions caused by heating in winter can be determined by subtracting the average constrained emissions" in winter [47]. This assumed that the majority of additional SO_2 concentrations during the heating seasons were due to heating. For the three years prior to 2017, when the performance targets on clean heating transition were assigned, the difference in SO_2 concentrations between the heating and non-heating seasons were different by as much as 3.85 times in Taiyuan, the capital city of Shanxi. For other pollutants, the differences between the heating and non-heating seasons were 1.77, 1.37, 1.72, and 1.26, for PM2.5, PM10, CO, and NO2, respectively. This again highlights the suitability of SO_2 as an indicator.

The study period was from January 2015 to March 2021. Generally, the central heating season runs for four months in North China, and 5–7 months in Northeast and Northwest China from October to April based on local temperatures. The 11 cities of Shanxi also have different heating periods based on their winter temperatures. The legal heating periods for Changzhi, Jincheng, Yuncheng, and Linfen runs from November 15 to March 15, Taiyuan,

Yangquan, Jinzhong, Lüliang, and Xinzhou from November 1 to March 31, and Datong and Shuozhou from October 25 to April 10. The heating periods differ between 4 months to 5 months, but the duration is also adjusted if the temperature drops earlier or if the low temperature continues even after the legal heating periods. In addition, individual heating in rural areas does not follow legal heating periods. Accordingly, in this study, we take the heating season from October to March as it is used in the Action Plans for Comprehensive Management of Air Pollution in the Beijing-Tianjin-Hebei and Surrounding Areas in the Autumn and Winter. Because the targets for clean heating transition were first put in February 2017 to treatment group 1 and in October 2018 to treatment group 2, SO₂ concentration data for 2015 and 2016 were used to examine changes before and after the targets were assigned.

The air pollutant data were acquired from the website of Chinese AQIstudy, which provides daily average levels of air pollutants from 367 Chinese cities. Weather data were obtained from the Chinese National Meteorological Information Center under the China Meteorological Administration.

4.3. Statistical Analysis

This study used a difference-in-differences (DID) approach to estimate the impact of the performance target on ambient SO_2 levels in Shanxi province. The DID approach is a widely used quasi-experimental design for the study of many policy questions, and it estimates the causal effect of policy interventions by comparing the change in outcome between treated and untreated controls before and after treatment. The DID model used in this study is represented by the following equations:

$$Y_{ct} = \beta_0 + \beta_1 T + \beta_2 D_c + \beta_3 (T \times D_c) + \beta_4 f(Z_{ct}) + \varepsilon_{ct}$$
(1)

where Y_{ct} is the daily average SO₂ concentration level for city c at time t, which was monitored before (T = 0) and after (T = 1) the treatment (D_c). β_3 is the coefficient of interest and estimates the average treatment effect on the treated samples. Z_{ct} includes a quadratic functional form of meteorological variables, temperature and humidity, and ε_{ct} represents residual errors.

$$\beta_3 = (E [Y_{ct} | c = treatment, t = 1] - E [Y_{ct} | c = treatment, t = 0]) - (E [Y_{ct} | c = control, t = 1] - E [Y_{ct} | c = control, t = 0])$$

$$(2)$$

This DID estimator compares the changes in SO_2 levels in the cities examined, to changes in SO_2 levels in control cities in Shanxi.

4.4. Document Analysis

This study also uses the document analysis method to examine the implementation process of a clean heating transition. It particularly focuses on the question of how the transition tasks were centralized, how Shanxi province responds or interprets the centralized tasks, why Shanxi has responded with an overzealous transition, how and what has supported the continuous transition efforts even though a gas shortage had occurred, and what the policy outcomes of inadequate heating supply are. To answer these questions, central and local policy documents and the statements of elite authorities were carefully examined. Moreover, the messages regarding clean heating transition written by the residents of Shanxi on the bulletin boards of Development and Reform Committee (DRC) of Shanxi were reviewed to understand the actual situation (at least partially), alongside examining environmental inspection reports that thoroughly describe the problems caused by a clean heating transition.

5. Results

5.1. Air Quality Improvement

Table 1 shows the regression results for Equation (1), which estimates the effect of air pollution controls on SO₂ levels. Column (1) begins by including only treatment assignment status (D), treatment period (T), and the DID estimator, which is defined as the interaction term of the two (T × D). The coefficient of interest β_3 is -13.709, which is statistically significant, indicating that the pollution controls reduced SO₂ concentration by approximately 14 µg/m³ (equivalent to a decrease of 21%), on average. The city-fixed effects are included to control for unobserved heterogeneity across the cities in column (2), and the results are robust.

Table 1. Baseline regression results.

	(1)	(2)	(3)	(4)
т	-29.571 ***	-29.571 ***	-27.079 ***	-1.962
1	(1.167) 6 122 ***	(1.161)	(1.002) 10 112 ***	(1.199) 7 376 ***
D	(1.062)		(0.913)	(0.812)
$\mathbf{T} imes \mathbf{D}$	-13.709 *** (1.353)	-13.815 *** (1.355)	-14.218 *** (1.162)	-15.700 *** (0.960)
Heating season			47.648 *** (0.514)	15.786 *** (1.166)
Temperature			()	0.589 *** (0.069)
Temperature ²				-0.027 *** (0.003)
Humidity				0.705 *** (0.065)
Humidity ²				-0.009 *** (0.001)
Wind speed				-10.214 *** (0.249)
City fixed effects	No	Yes	No	No
K ² Observation	0.161 24 110	0.157 24 110	0.382 24 110	0.600 24 106
Observation	24,110	24,110	24,110	24,106

*** p < 0.01 ** p < 0.05 * p < 0.10.

In column (3), the heating season is included to control for coal-fired heating during winter. The heating season is typically between October to March of the following year, during which time the ambient air pollution levels are at their highest. The result in column (3) suggests that the SO₂ level is 47.65 μ g/m³ (approximately 205%) higher during the heating season compared to other months. Column (4) includes weather controls such as wind speed and quadratic forms of temperature and relative humidity. The results show that the clean heating transition reduced SO₂ levels by approximately 16 μ g/m³, (equivalent to a decrease of 23.74%), providing empirical evidence for air pollution reduction due to the transition to clean energy.

Furthermore, the treatment group is now split into two groups: the 2 + 26 key cities and the Fen-Wei Plain region in order to examine the effect of performance targeting. Table 2 shows the heterogeneous treatment effects of the two groups. D1 and D2 represent the Fen-Wei Plain region and the 2 + 26 key cities, respectively. In columns (1)–(4), we find a greater reduction in SO₂ levels in the cities of the Fen-Wei Plain than in the 2 + 26 key cities, indicating that targeting achievement goals did not result in better outcomes. We also tested whether the treatment effects are significantly different from each other, and the large F statistic suggests that we can reject the equality hypothesis at the 1% level.

Table 3 shows the regression results for the heating seasons only, and the indicator for the heating season is now omitted from the regressions. In all model specifications, the clean heating transition led to a significant reduction in SO_2 levels in winter, and the larger impacts were found in the Fen-Wei Plain region.

	(1)	(2)	(3)	(4)
	-29.571 ***	-29.571 ***	-27.077 ***	3.688 ***
1	(1.167)	(1.160)	(1.002)	(1.258)
D1	6.675 ***		11.123 ***	12.922 ***
DI	(1.132)		(0.973)	(0.901)
D2	5.216 ***		8.461 ***	1.950 **
D2	(1.240)		(1.065)	(0.898)
$T \times D1$	-15.562 ***	-15.562 ***	-16.632 ***	-19.498 ***
$1 \wedge D1$	(1.523)	(1.515)	(1.308)	(1.126)
$T \times D^2$	-11.959 ***	-11.957 ***	-11.653 ***	-11.008 ***
$1 \wedge D2$	(1.543)	(1.535)	(1.325)	(1.064)
Heating season			47.674 ***	15.574 ***
Treating beasers			(0.514)	(1.161)
Temperature				0.573 ***
r				(0.069)
Temperature ²				-0.028 ***
Temp eruture				(0.003)
Humidity				0.649 ***
5				(0.065)
Humidity ²				-0.008 ***
5				(0.001)
Wind speed				-10.254
City fine d offerst	NL	Vee	N.	(0.248)
E statistic (n seeless)	INO (EE (0.010E)	1es	INO 1(07(0,0000)	INO
r statistic (<i>p</i> -value)	0.05 (0.0105)	0.03 (0.0100)	10.97 (0.0000)	0.000000
K ²	0.162	0.157	0.382	0.604
Observation	24,110	24,110	24,110	24,106

 Table 2. Regression results: Two treatment groups.

_

*** p < 0.01 ** p < 0.05 * p < 0.10.

 Table 3. Regression results: Heating season only.

	(1)	(2)	(3)
т	-46.569 ***	-46.603 ***	8.457 ***
1	(2.075)	(2.037)	(2.861)
D1	28.561 ***		27.996 ***
DI	(2.044)		(2.004)
D2	19.688 ***		4.017 **
D2	(2.234)		(2.036)
$T \times D1$	-36.062 ***	-36.028 ***	-39.017 ***
$1 \times D1$	(2.845)	(2.793)	(2.564)
$T \times D^2$	-25.948 ***	-25.992 ***	-22.314 ***
$1 \times D2$	(2.851)	(2.799)	(2.414)
Temperature			0.313 **
Temperature			(0.125)
Tomporaturo ²			-0.055 ***
lemperature			(0.010)
Humidity			0.994 ***
Trainfarty			(0.157)
Humidity ²			-0.012 ***
Turnany			(0.001)
Wind speed			-16.321 ***
Wild Speed			(0.507)
City fixed effect	No	Yes	No
F statistic (<i>p</i> -value)	13.44 (0.0002)	6.63 (0.0100)	16.97 (0.0000)
R^2	0.299	0.288	0.521
Observation	9470	9470	9467
*** ·· · 0 01 ** ·· · 0 0E * ·· · 0 10			

*** p < 0.01 ** p < 0.05 * p < 0.10.

5.2. Rapid and Large-Scale Transition

As the APACCP 2013–2017 entered a critical stage, the central government responded by instituting more detailed measures and increasing target levels to induce local implementers to work in the interests of the central authorities. When local governments appeared to face difficulties following the preferences of the center, the central government routinely reacted in an opposing manner [48], so the ad hoc campaign-style implementation was repeated at the end year of APACCP 2013–2017. In addition, as the authorities sent clearer messages on the importance of environmental protection and the environmental inspection system was strengthened by the central leadership, the tasks of the clean heating transition were highly politicized.

In February 2017, the Chinese government introduced the 2 + 26 key cities as the air pollution transmission channel, and mandated each city to complete replacing coal with gas or electricity for a minimum of 0.05 million to a maximum of 0.01 million households by the end of October 2017 through WPACCP 2017. The total ranged from a minimum of 1.4 million to a maximum of 2.8 million households. Soon after August of the same year, the APCMAP 2017–2018 was issued with upgraded targets and detailed requirements for clean heating transition. The 2 + 26 key cities were asked to complete the relevant transformation in a total of 3.55 million households by October 2017. Unlike the WPACCP 2017, the APCMAP 2017–2018 assigned differential targets to the 28 cities. Consequently, the transition targets in total were increased from 0.75 million to as much as 2.15 million households in 6 months, but the deadline to complete the transition was only two months for all the 2 + 26 key cities.

The top-tier authorities, highlighting the significance of the clean heating transition and air pollution control, provided stronger incentives to vigorously promote coal-to-gas and coal-to-electricity projects. At the 14th meeting of the Central Financial and Economic Leading Group in October 2017, the importance of the clean heating transition in preventing winter haze and ensuring a warm winter for the people was highlighted by President Xi Jinping [49]. At the 19th National Congress of the CPC, the President stressed pollution prevention and control as one of the "three tough battles" to be fought and won by 2020 to "finish building a moderately prosperous society in all respects and achieve the first centenary goal" [50]. President Xi highlighted the (i.e., environmental protection, green and ecosystem) more than the economy in his speech; each was mentioned 89 and 70 times, respectively. This was a reversal compared to the speech delivered by former President Hu Jintao, who used 104 and 74 words related to the economy and environment respectively at the CPC 18th National Congress in 2012 [51]. Additionally, at the National People's Congress held in March 2017, Premier Li Keqiang mentioned that the prevalence of clean winter heating, through the replacement of coal with gas or electricity, should be more than 3 million households [52]. The announcement by Premier Li was made only a few days after the WPACCP 2017 was issued with the transition target from a minimum of 1.4 to a maximum of 2.8 million households. As he listed the clean heating transition tasks firstly to reduce coal burning and environmental protection, coal replacement became the center of environmental issues.

The new central environmental inspection system has also clearly specified the attitude of the central government towards environmental protection. To address the lack of accountability and control over local governments, the central government under Xi Jinping established the central inspection team with support from the two most powerful CCP institutions: the Organization Department and the Central Commission for Discipline Inspection [40,53]. The Organization Department is responsible for party personnel assignments, so it determines the general regulations of the cadre performance evaluation system and assesses cadre performance accordingly [15]. The Central Commission for Discipline Inspection is the most powerful anti-corruption organ in China. The process was highly politicized because the officials from the two organs and retired ministry officials joined the inspection team and the focus of the inspection was adjusted to the highest-ranking local party and government leaders instead of polluting enterprises. Under this strengthened control system, any local leader who was observed to be involved in misconduct or as inattentive to environmental protection, could face immediate demotion or removal from the positions and face possible corruption investigation regarding protectionist behavior [40].

Alongside the central environmental inspection, the MEP organized another inspection system under the Enhanced Supervision Plan for Air Pollution Prevention and Control for the Beijing-Tianjin-Hebei and Surrounding Areas in 2017–2018. For the clean heating transition, the 11th to 25th rounds of the inspection were systemically assigned for APCMAP 2017–2018. The inspection was divided into two stages. The first stage was to ensure and urge all the listed and time-limited tasks to be completed at this stage, so the inspection teams were dispatched to each of the 2 + 26 key cities every month from 1st September to 9th November, 2017. The second stage was conducted using "two-way feedback" inspection system from November 10, 2017 to March 29, 2018 [54]. To improve the efficiency of regulation and address the problem of inadequate enforcement, the MEP planned to use satellite remote sensing and monitor heavily polluted areas. The 2 + 26key cities were divided into 3 km \times 3 km grids, and 3600 highly polluted grids were selected for intensive monitoring among 36,783 grids [55]. Once a high concentration of air pollution was detected by the sensor, the inspection team and the environmental protection department of each city conducted investigations, and the local governments belonging to the particular grids were required to report on the reasons and problems of the pollutant increase within the same day when it was detected. The investigations were carried out until the grids were significantly improved. Through strengthening of the inspection systems, the central government gave clear signals to local governments to prioritize environmental protection, and Shanxi responded by exceeding the planned targets.

In addition to these strong political pressures, the wave of dismissals of local leaders who had failed to reach environmental targets during the 12th Five-Year Plan periods had increased the fear of failure, which could undeniably damage their careers [56]. In addition, clean heating conversion projects could increase the employment rate, which has always been regarded as a strategic task for the political and performance legitimacy of the CPC. According to a report, job-seeking day laborers were employed at the construction sites because infrastructure buildings and low-carbon equipment installation for clean heating required an intensive workforce. Moreover, coal switching projects could induce immediate work performance. If they use less coal, especially scattered coal for individual heating, pollutant emissions can be reduced instantly. Local leaders did not have to hesitate; instead, they were motivated to respond by reinforcing actions.

The Shanxi province administration perceived changes in the interests of the central government in environmental protection, and responded overzealously on transition tasks. Shanxi received a transition target of 0.39 million households from the central government in 2017. However, the provincial government assigned transition targets to the rest of its seven centrally untargeted, prefecture-level cities through the APCMAP 2017–2018 of the Shanxi province. Therefore, the total transition target set in Shanxi was 1.014 million households which was 2.6 times higher than what the central government had assigned for the province (Table 4). As a result, 1.13 million households completed the transition in 2017. This means that the performance target assigned by the central government was surpassed by nearly 190%. Particularly, among the 1.13 million households, only 0.42 million households completed the coal-to-gas or coal-to-electricity transitions in the targeted, 2 + 26 key cities group [57]. This indicates that the transition efforts were not concentrated in the 2 + 26 key cities group, but were fairly distributed in untargeted cities, as was planned by Shanxi provincial government (Table 4).

			(Unit: Million Households)						
	Central Gove	rnment Plan	Provincial Government Plan	Prefecture-Level City Government Plan					
Taiyuan		0.112	0.112	0.134					
Yangquan	0.200	0.056	0.058	0.058					
Changzhi	0.390	0.118	0.118	0.118					
Jincheng		0.112	0.111	0.100					
Jinzhong			0.100	0.111					
Yuncheng			0.063	0.064					
Linfen			0.191	0.100					
Lüliang			0.100	0.102					
Datong			0.053	0.053					
Xinzhou			0.059	0.059					
Shuozhou			0.050	0.050					
Total	0.390	0.398	1.014	0.948					

Table 4. Shanxi province's clean heating transition targets at the three levels of government.

Source: Central Government Plan [28], Provincial Government Plan [44], Prefecture-level City Government Plan [58–68].

6. Discussion

6.1. Clean Winter in Shanxi Province

The SO₂ concentrations in Shanxi were continuously reduced both annually and every heating season, and have not reverted to the previous levels since the performance targets for clean heating transition were imposed in 2017. The annual mean of SO₂ concentrations in Shanxi fell under China's first-level concentration limit of 20 μ g/m³, and the maximum monthly concentrations also fell sharply to 40 μ g/m³ in 2020 (Table 5). Therefore, the mean concentrations of SO₂ during October 2020 to March 2021 were only 1.8 times that of those in spring and summer (April 2020–September 2020) of the same year (Table 6).

Compared to 2016, when the annual mean concentrations of SO₂ were 60 μ g/m³ and the mean concentrations in autumn and winter were 4.0 times higher than those in spring and summer (28 μ g/m³–113 μ g/m³), the annual and heating season's SO₂ concentrations in 2020 were reduced by one-third (60 μ g/m³–18 μ g/m³) and controlled by 81% (113 μ g/m³–22 μ g/m³). As a result, the targets set for Shanxi from the APCMAP 2017–2018 and the TYAP 2018–2020 were both met. The SO₂ concentration in Shanxi during the 2017–2018 heating season was 56 μ g/m³. As it was reduced by 50% from 113 μ g/m³ in the previous heating season, it surpassed the 40% reduction target. The TAYP 2018–2020 target of 50% reduction by 2020 compared to 2015 was 18 μ g/m³ from 56 μ g/m³. This was reduced by 68% (Table 6).

The reduction in SO₂ concentrations in the Fen-Wei Plain region was greater than that in the 2 + 26 key cities during the study period. SO₂ concentrations in all three groups declined further during the first assessment of the heating season. In particular, the SO₂ concentration in the 2 + 26 key cities decreased the most. SO₂ concentrations in the other two regions (Fen-Wei Plain region and control group), for which the central government did not specify a target amount, also reduced significantly between October 2017 to March 2018 (Table 6). In the subsequent heating seasons (October 2018–March 2019 and October 2019–March 2020), the SO₂ levels in the Fen-Wei Plain showed a continuous decrease and were more significant than those in the 2 + 26 key cities. A higher transition goal was set in the Fen-Wei Plain group than in the 2 + 26 key cities since 2018; this might have resulted in a greater reduction in SO₂ concentration in the Fen-Wei group (Figure 3).

	(Unit: $\mu g/m^3$)											
	2 + 26 Key Cities Group (4) Fen-Wei Plain Group (4) Control Group (3)						(3)					
	Taiyuan	Yangquan	Changzhi	Jincheng	Jinzhong	Yuncheng	Linfen	Lüliang	Datong	Xinzhou	Shuozhou	Average
2015	69	59	49	56	72	54	64	69	42	57	76	56
2015 *	179	126	133	115	177	130	260	150	80	114	187	150
2016	68	62	61	70	88	66	83	63	48	49	67	60
2016 *	186	118	164	139	263	154	351	191	130	120	154	179
2017	55	50	43	48	84	52	79	69	45	49	47	52
2017 *	163	115	145	99	249	113	311	184	101	116	143	158
2018	27	30	20	23	34	27	43	37	29	31	33	28
2018 *	63	62	48	47	91	83	116	91	50	66	70	72
2019	22	23	16	16	26	15	28	29	30	29	31	22
2019 *	58	65	41	40	68	41	107	91	65	65	92	67
2020	17	20	17	13	20	13	19	18	30	20	26	18
2020 *	33	37	33	24	44	21	54	31	52	40	76	40

Table 5. The annual mean and maximum monthly concentrations of SO₂ in Shanxi province.

Source: Calculated by authors using data from AQIstudy. Year with * represents maximum monthly value in each year. Numbers in parentheses indicate number of cities belong each group.

(Unit: µg/m ³)												
	2 + 26 Key Cities Group (4) Fen-Wei Plain Group (4) Control						Group (3)					
	Taiyuan	Yangquan	Changzhi	Jincheng	Jinzhong	Yuncheng	Linfen	Lüliang	Datong	Xinzhou	Shuozhou	Average
April 2015–September 2015	28	32	28	39	36	35	24	41	24	37	35	30
October 2015–March 2016	91	94	97	93	101	102	53	75	63	62	109	85
April 2016–September 2016	29	34	17	37	37	34	35	27	24	32	32	28
October 2016–March 2017	121	84	101	85	184	87	196	127	81	85	94	113
April 2017–September 2017	25	28	16	29	35	26	34	30	26	28	19	25
October 2017-March 2018	52	54	33	43	75	61	79	74	42	52	52	56
April 2018–September 2018	13	17	10	15	14	13	20	17	18	19	15	14
October 2018-March 2019	35	37	26	25	43	25	51	52	42	42	53	39
April 2019–September 2019	13	14	11	12	13	9	14	17	20	22	16	13
October 2019-March 2020	25	23	19	17	30	16	27	25	37	29	41	26
April 2020-September 2020	11	14	13	10	11	10	11	15	21	16	15	12
October 2020-March 2021	21	26	20	15	29	15	20	19	35	20	24	22

Table 6. Shanxi province's SO₂ mean concentrations: Spring and summer vs. autumn and winter.

Source: Calculated by authors using data from AQIstudy. Numbers in parentheses indicate number of cities belong each group.



Figure 3. Policy targets and transition achievements in Shanxi. Source: 2 + 26 key cities group [28,69–71]; Fen-Wei Plain region group [62–65,71–73] The Fen-Wei Plain region group's 2017 target data is from each prefecture-level city's APCMAP 2017–2018. The unit is one million households.

The groups of the 2 + 26 key cities and the Fen-Wei Plain region both exceeded the transition coverage rate set by the central government. The Clean Winter Heating Plan in Northern China (2017–2021) set targets for clean heating rates to reach 50% in northern China by 2019 and 70% by 2021. However, higher targets were assigned to the 2 + 26 key cities: 90% and above should be attained in urban areas, 70% in urban fringe areas, and 40% in rural areas by 2019. Furthermore, the transition rates should be 100%, 80%, and 60% by 2021. However, all of the 11 urban cities in Shanxi already attained 98.81% of transition coverage rate, and its urban fringe areas also achieved 86.46% by 2018. The total coverage rate in the rural areas of Shanxi reached only 31.59% in 2018, but rural areas in the 2 + 26 key cities and the Fen-Wei Plain region was 40.96%, so these two regions were ahead of government plans in 2018 (Table 7). However, this unexpected excess disrupted the natural gas supply and forced people to endure cold winters.

Table 7. 2018 Shanxi province's 11 cities clean heating coverage rate.

	Urban Areas	Urban Fringe Areas	Rural Areas	Total
Taiyuan	100%	97.81%	71.98%	96.86%
Yangquan	100%	84%	37%	76.61%
Changzhi	100%	94%	31%	69.45%
Jincheng	96.10%	92.20%	35.98%	57.13%
Jinzhong	100%	88.25%	29.36%	70.86%
Yuncheng	100%	82.77%	49.13%	71.93%
Linfen	96.40%	91.16%	43.81%	70.48%
Lüliang	100%	98.37%	29.42%	74.67%
Datong	93.82%	59.25%	11.28%	50.01%
Xinzhou	94.11%	81.53%	9.67%	59.93%
Shuozhou	99.90%	85.71%	6.32%	57.19%
Total	98.81%	86.46%	31.59%	71.75%

Source: Investigation report on clean heating work [74].

The central government had slowed the clean heating transition through emergency measures because the gas shortages prevented sufficient heating; however, the efforts towards the clean heating transition were not retreated after little adjustment in 2018.

This study finds that the following two factors supported continuous transition efforts: firstly, after the end of APACCP 2013–2017, the Chinese government continuously took serious air pollution control and prevention by issuing the TYAP 2018–2020. However, after experiencing the disastrous social consequences of not securing winter heating for residents, the central government did not include clean heating transition targets in the TYAP 2018–2020 and also did not assign the targets for the 2 + 26 key cities and the Fen-Wei Plain region through the APCMAP. Instead, local governments set transition targets that have been included in the annual APCMAP since 2018. Nevertheless, Vice-premier, Han Zheng who is the leader of the Leading Small Group for APACCP in the Beijing-Tianjin-Hebei and Surrounding Areas and the Ministers of MEE, on a variety of occasions, have reiterated that President Xi attaches great importance to clean heating in winter. In addition, the 2 + 26 key cities and the Fen-Wei Plain region have been centered on the central environmental inspection. Therefore, the transition efforts did not lose its intensity, and the air quality could continue to improve.

Secondly, the Chinese government did not modify the competitive funds that incentivized local governments to continue the fast transition. In 2017, the Ministry of Finance (MOF), MEP, Ministry of Housing and Urban-Rural Development (MHURD), and National Energy Agency (NEA) created a government subsidy policy for clean heating transition, but it was announced that it would be effective only for three years. Three batches of 43 cities were selected to receive the competitive funds for three years. Twelve cities were selected in the first year, 2017. In 2018, 23 more cities began to be funded. Eight cities were added in 2019. All cities in the 2 + 26 key cities and the Fen-Wei Plain region, except Beijing, received this special fund, but the amounts were different based on levels of cities. It proposed that municipalities, provincial capital cities, and prefecture-level cities of the 2 + 26 key cities must be granted 1 billion CNY, 0.7 billion CNY, and 0.5 billion CNY respectively, and cities in Fen-Wei Plain were to receive 0.3 billion CNY regardless of its type [75,76].

The local governments had to achieve a high rate of transition, while funds from the central government supported them. Otherwise, more financial pressure would be imposed. Even though the central government reversed the 'coal ban' attitude for heating systems in December 2017, the local government aimed for a clean heating transition as the special funds were continuously provided. Therefore, the transition efforts were continued in Shanxi as evidenced by the complete transition of 1.13 million, 0.962 million, and 1.43 million households in 2017, 2018, and 2019, respectively [77–79]. According to Shanxi province clean heating work leading group in January 2021, the province completed a total of 5.0945 million clean heating transitions from 2017 to 2020, and 1.5745 million households completed the coal switch in 2020. Consequently, even though the clean heating coverage of the province had already reached more than 70% in 2018 (Table 7), Shanxi had continuously put radical efforts on clean heating transition as the coverage rate reached more than 80% in 2020 [80].

6.2. Cold Winter in Shanxi Province

The rapid and massive clean heating transition cleaned the air, but it failed to secure sufficient heating for Shanxi residents. As a result, Shanxi residents had to endure cold winters, and the central government had to reverse its ambitious plans for coal ban and promoting gas plants. This undermined the energy transition efforts. Firstly, Shanxi residents had to spend cold winters because severe gas shortages and inadequate infrastructure prevented sufficient winter heating. Even though the Shanxi DRC limited gas supplies for the non-residential sectors including industry, commerce, hotels, and offices, it was frequently reported that the gas was often cut off, supplied only a few hours daily, or not restored for longer than 24 h due to a supply shortage [81–83].

Throughout "looking back (huitoukan)" Rectification Plan of Shanxi on the central and the MEE's inspection, the problems of low gas flow, pipe installation for the last meter, and insufficient subsides that hindered the transition process were exposed. During 2017–2018, 0.8859 million households completed the coal-to-gas switch in Shanxi, but only 0.6133 million households could receive gas. Therefore, the overall rate of gas flow was 70%, but there were wide variations among the cities. The rates of gas flow in the cities of Yuncheng (Fen-Wei Plain region), Datong (control group), and Shuozhou (control group) were only 20.3%, 3.1%, and 44.8%, respectively. Due to uneven subsidy standards in Yangquan (2 + 26 key cities), it was reported that some counties inevitably went back to coal heating even though they were connected to pipelines. Moreover, it revealed that only one of the 24 gas companies in Changzhi city (2 + 26 key cities) received government subsidies. The remaining 23 gas companies were unable to sell gas to the residents at a high market price, so they had to endure difficult business conditions until subsides were provided by the central governments [84].

The DRC consultation bulletin boards of Shanxi contained complaints and questions about the last meter pipes, improperly distributed subsides, and high costs of heating from residents all over Shanxi, but especially from Changzhi, Linfen, and Jinzhong. Bulletin boards cannot represent the overall situation, but they have hinted at the partial ambiance of the transition process. Many criticized the pipe networks laid out at the door of their houses, but only those who paid for the installation fees for the last meter pipes could be connected to pipelines and were supplied gas for heating. Others even complained that they were threatened to pay for the installation as early possible and were informed that the price would be higher next year. Although the Shanxi provincial government issued a prohibition on charging construction and installation fees to residents at the end of October 2017, the brutality of the concerned companies continued at least until 2019, when the complaints of the residents were found on the bulletin board [85].

Secondly, the central government urgently revised the clean heating transition measures to address inadequate heating supply and reversed its ambitious plans for energy transition. On the official starting day (November 15, 2017) of central heating in northern China, the NDRC pledged an overall affluent gas supply for the 2017–2018 heating season. Its confidence in smooth and sound gas supply was disregarded 20 days later when the NDRC, NEA, MHURD, and MEP issued emergency measures to secure gas supply and demanded that the coal switch should be pursued once natural gas supply is secured [86]. The emergency measures included an increase in domestic production and LNG imports, diverting natural gas for industrial and power generation to residential use, and transferring more gas from the southern region to the northern heating zone [56]. Consequently, a study found that pollution was highly likely to be transferred to the south as they had to use more coal to replace gas [9].

The outcomes of the APACCP 2013-2017 not only caused gas shortages but also changed the government's attitude towards gas deployment. The NDRC issued two documents on gas deployment during the 13th Five-Year Plan period (2016–2020). The 13th Five Year Plan for Natural Gas Development set the proportion of natural gas in the primary energy consumption structure to attain 8.3-10% and 110 GW of gas-fired generation capacity by 2020. The Opinions on Speeding up Natural Gas Utilization documents the upgradation in the share of natural gas in primary energy consumption to 10%. - To increase gas consumption, both stressed the need to develop gas peaking plants, combined heat and power plants, and promote gas plants to supplement renewable production. Since the central governments reversed tones in the policies, the driving force on the promoting gas plants immediately faded away as they only asked for the development of gas peaking plants and limited new combined heat and power plants in plans and measures on air pollution and clean heating transitions such as the TYAP 2018–2020 and the APCMAP 2018–2019 issued in later years [87]. Moreover, Premier Li Keqiang reiterated at the 2019 National Energy Commission Meeting that the provinces should utilize coal, gas, or electricity according to local conditions to ensure winter heating supply. This was in parallel to previous emergency measures that cautioned against a "one-size fits all" approach on the clean heating transition to slow down the coal to gas or electricity

conversion. In conclusion, the proportion of natural gas and capacity of coal-fired plants was recorded at only 8.4% and 98 GW at the end of 2020, falling short of targets [5].

7. Conclusions

Despite the implementation of the APACCP 2013–2017, the red alerts declared in December 2015 and December 2016 in Beijing represented the severity of air quality problems in winter. As a result, heating started to receive special attention as a major contributor to air pollution in winter, and President Xi Jinping emphasized the importance of a clean heating transition in December 2016. In 2017, the deadline year for the APACCP 2013–2017, President Xi and Premier Li reiterated the importance of environmental protection, a series of measures targeting the clean heating transition were enhanced, and environmental inspection was strengthened by central leadership. Therefore, the clean heating transition tasks were highly centralized.

Since the Chinese government has assigned performance targets to environmental and energy concerns under the cadre responsibility and evaluation system to enhance policy implementation, questions on the effectiveness of target-based implementation have been centered on studies of China's environmental governance. Because environmental performance had been added to the cadre evaluation criteria, theoretically, the cadre responsibility and evaluation system would have incentivized local cadres and officials to prioritize environmental policies. However, the existing literature underlines that the target-based approach creates an implementation gap due to poor incentives and structural shortcomings. Therefore, these studies eventually claim that the central government should have a clearer intention of environmental protection and an enhanced inspection system to narrow the gap and successfully achieve the intended goals.

This study sought to contribute to the lack of empirical evidence on the impact of the recentralized governance on environmental performance by examining the implementation of a clean heating transition. It investigated the impact of a centralized clean heating transition for SO₂ in Shanxi Province from January 2015 to March 2021. This study showed that the SO₂ concentration in Shanxi increased until 2016, then sharply decreased in 2017 when the clean heating transition was vigorously promoted. The clean heating transition led to a significant reduction in SO_2 levels in winter, and larger impacts were found in the Fen-Wei Plain region. It is noteworthy that the difference in SO₂ concentrations between the heating and non-heating seasons in Shanxi had decreased from four times in 2016 to 1.8 times in 2020, and the contribution of the clean heating transition must have been significant. It continued to decrease and attained the national level 1 of the annual mean SO_2 concentration of 20 $\mu g/m^3$ in 2020. This study showed that the annual target of air pollution reduction and clean heating transition through APCMAP and three-year competitive funds have supported the transition efforts continuously. Therefore, it is understood that the SO₂ concentration of Shanxi had a greater effect over the past four years (2017–2020), but not from 2013 when the APACCP 2013–2017 began. Moreover, it can be evaluated that the SO₂ concentration reduction effort was successful based on the fact that the SO₂ reduction target was exempted from the 14th Five-year Plan while it had been specified throughout the last 15 years, from the 11th Five-year Plan to the 13th Five-Year Plan.

The administration of Shanxi province perceived changes in the tone and interest of the central government towards environmental protection and invested overzealously on transition tasks. The central government assigned a transition target of 0.39 million households in 2017. However, the provincial government assigned transition targets to the rest of its centrally untargeted cities and completed the transition in a total of 1.13 million households. It surpassed the transition target assigned by the central government by nearly 190%. Interestingly, only 0.42 million households completed the transition in the 2 + 26 key cities, indicating that the transition efforts were fairly distributed in Shanxi as was planned. Further investigation is required to investigate the motivations behind the over-achievement of Shanxi. However, at this stage, it is assumed that the fear of failure to achieve performance targets and the willingness to impress higher authorities promoted transition rates that exceeded the planned targets.

Unfortunately, the rapid and large-scale transition cleaned the winter air, but failed to secure sufficient heating for the residents of Shanxi; they had to endure the cold winters. These adverse consequences impacted the national energy transition plan. The central government had to urgently adjust clean heating transition measures to address inadequate heating supply and reverse its ambitious plans on coal bans and promoting gas plants. This implies that recentralization may have been able to provide proper incentives to prioritize the environmental issues as much as economic growth, but it could not control the autonomy of how local governments and how it achieved the central targets. As a result, recentralization was not as effective as theoretically anticipated and undermined energy transition efforts.

In China, the central government announced the policies and the local governments implemented them. If this resulted in adverse results, emergency documents were issued to correct the original plans. For the clean heating transition, an identical cycle was repeated. However, the insufficient heating supply caused by the gas shortage was far from sufficient to be resolved by urgent documents with a message of easing the coal ban. To speed up the conversion and encourage local residents to participate in transition tasks, the coal stoves were removed before the clean heating devices were properly installed. Therefore, even if coal re-combustion was allowed, heating could not be supplied. Since the recentralization of environmental governance could not control the autonomy of local governments, the cadre responsibility and evaluation system should prepare new incentives, punishment or revision of the cadre evaluation indicators to prevent the implementation process from being distorted.

Author Contributions: Conceptualization, J.-S.L., E.J.L. and M.J.K.; methodology, J.-S.L., E.J.L. and M.J.K.; software, M.J.K.; validation, E.J.L. and M.J.K.; formal analysis, E.J.L.; data curation, E.J.L.; writing—original draft preparation, E.J.L.; writing—review and editing, J.-S.L., M.J.K.; visualization, E.J.L.; supervision, J.-S.L., M.J.K. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data used to support the results of this study can be found here: https://www.aqistudy.cn/historydata/ (accessed on 5 April 2021); http://www.nmic.cn/ dataService/cdcindex/datacode/SURF_CLI_CHN_MUL_DAY_140/show_value/normal.html; (accessed on 8 April 2021); https://data.stats.gov.cn/easyquery.htm?cn=C01 (accessed on 13 May 2021) https://www.yearbookchina.com/navibooklist-n3020013309-1.html (accessed on 13 May 2021).

Acknowledgments: This study was supported by the KU-DIS Special Research Grant (2020).

Conflicts of Interest: The authors declare no conflict of interest.

References

- Wang, Z.; Li, C.; Cui, C.; Liu, H.; Cai, B. Cleaner Heating Choices in Northern Rural China: Household Factors and the Dual Substitution Policy. J. Environ. Manag. 2019, 249, 109433. [CrossRef] [PubMed]
- Yuan, M.; Thellufsen, J.Z.; Lund, H.; Liang, Y. The First Feasible Step towards Clean Heating Transition in Urban Agglomeration: A Case Study of Beijing-Tianjin-Hebei Region. *Energy Convers. Manag.* 2020, 223, 113282. [CrossRef]
- 3. China: Development of National Heat Pricing and Billing Policy. Available online: https://openknowledge.worldbank.org/ handle/10986/17908 (accessed on 2 April 2021).
- The National Development and Reform Commission of the People's Republic of China. Clean Winter Heating Plan in Northern China (2017–2021). Available online: https://www.ndrc.gov.cn/xxgk/zcfb/tz/201712/t20171220_962623.html?code=&state=123 (accessed on 19 April 2021).
- 5. The National Bureau of Statistics of China. China Statistical Yearbook 2020. Available online: http://www.stats.gov.cn/tjsj/ndsj/ 2020/indexeh.htm (accessed on 14 May 2021).

- 6. Shi, Y.; Xi, Z.; Simayi, M.; Li, J.; Xie, S. Scattered Coal Is the Largest Source of Ambient Volatile Organic Compounds during the Heating Season in Beijing. *Atmos. Chem. Phys.* **2020**, *20*, 9351–9369. [CrossRef]
- 7. Werner, S.; Gong, M. District Heating Research in China; Svensk Fjärrvärme AB: Stockholm, Sweden, 2014.
- Liu, L.; Fu, L.; Jiang, Y.; Guo, S. Major Issues and Solutions in the Heat-Metering Reform in China. *Renew. Sustain. Energy Rev.* 2011, 15, 673–680. [CrossRef]
- Wang, S.; Su, H.; Chen, C.; Tao, W.; Streets, D.G.; Lu, Z.; Zheng, B.; Carmichael, G.R.; Lelieveld, J.; Pöschl, U.; et al. Natural Gas Shortages during the "Coal-to-Gas" Transition in China Have Caused a Large Redistribution of Air Pollution in Winter 2017. Proc. Natl. Acad. Sci. USA 2020, 117, 31018–31025. [CrossRef] [PubMed]
- 10. Xinhua News. China Declares War Against Pollution. Available online: http://www.npc.gov.cn/zgrdw/englishnpc/Special_12 _____2/2014-03/06/content_1839511.htm (accessed on 1 July 2021).
- 11. Li, K. Report on the Work of the Government (2014). 2014. Available online: http://english.www.gov.cn/archive/publications/ 2014/08/23/content_281474982987826.htm (accessed on 1 July 2021).
- 12. Beijing Announces Emegency Plan for Heavy Air Pollution. Available online: http://www.cma.gov.cn/2011xwzx/2011xmtjj/20 1310/t20131022_229458.html (accessed on 15 May 2021).
- Chen, Z.; Chen, D.; Wen, W.; Zhuang, Y.; Kwan, M.-P.; Chen, B.; Zhao, B.; Yang, L.; Gao, B.; Li, R.; et al. Evaluating the "2 + 26" Regional Strategy for Air Quality Improvement during Two Air Pollution Alerts in Beijing: Variations in PM2.5 Concentrations, Source Apportionment, and the Relative Contribution of Local Emission and Regional Transport. *Atmos. Chem. Phys.* 2019, 19, 6879–6891. [CrossRef]
- 14. Xinhua News. The 14th Meeting of the Central Financial Leading Group Was Held. Available online: http://www.gov.cn/xinwen/2016-12/21/content_5151201.htm (accessed on 23 May 2021).
- 15. Ran, R. Perverse Incentive Structure and Policy Implementation Gap in China's Local Environmental Politics. *J. Environ. Policy Plan.* **2013**, *15*, 17–39. [CrossRef]
- Eaton, S.; Kostka, G. Does Cadre Turnover Help or Hinder China's Green Rise? Evidence from Shanxi Province. In *Chinese Environmental Governance*; Ren, B., Shou, H., Eds.; Palgrave Macmillan US: New York, NY, USA, 2013; pp. 83–110. ISBN 978-1-349-46853-9.
- 17. Eaton, S.; Kostka, G. Central Protectionism in China: The "Central SOE Problem" in Environmental Governance. *China Q.* 2017, 231, 685–704. [CrossRef]
- 18. Kostka, G.; Nahm, J. Central–Local Relations: Recentralization and Environmental Governance in China. *China Q.* **2017**, 231, 567–582. [CrossRef]
- 19. Wang, A.L. The Search for Sustainable Legitimacy: Environmental Law and Bureaucracy in China. *Harv. Environ. Law Rev.* 2013, 37, 365–440. [CrossRef]
- 20. Hsu, A. Limitations and Challenges of Provincial Environmental Protection Bureaus in China's Environmental Data Monitoring, Reporting and Verification. *Environ. Pract.* 2013, 15, 280–292. [CrossRef]
- 21. Lo, K. China's Low-carbon City Initiatives: The Implementation Gap and the Limits of the Target Responsibility System. *Habitat Int.* **2014**, *42*, 236–244. [CrossRef]
- 22. Kostka, G. Command without Control: The Case of China's Environmental Target System. Regul. Gov. 2016, 10, 58–74. [CrossRef]
- 23. Ran, R. Understanding Blame Politics in China's Decentralized System of Environmental Governance: Actors, Strategies and Context. *China Q.* 2017, 231, 634–661. [CrossRef]
- 24. Xu, T. The Selective Centralization in Decentralization: China's Environmental Vertical Management Reform with a Case Study in Hebei Province. *Int. J. Water Resour. Dev.* **2021**, 1–24. [CrossRef]
- 25. The State Council of the People's Republic of China. The Action Plan for Air Pollution Prevention and Control. Available online: http://www.gov.cn/zhengce/content/2013-09/13/content_4561.htm (accessed on 13 June 2021).
- The State Council of the People's Republic of China. Measures to Evaluate the Implementation of the Action Plan for Air Pollution Prevention and Control. Available online: http://www.gov.cn/zhengce/content/2014-05/27/content_8830.htm (accessed on 13 June 2021).
- The Ministry of Environmental Protection of the People's Republic of China. The Work Plan for Air Pollution Prevention and Control for Beijing-Tianjin-Hebei Region and Surrounding Areas in 2017. Available online: https://www.mee.gov.cn/dtxx/2017 03/t20170323_408663.shtml (accessed on 11 May 2021).
- 28. The Ministry of Environmental Protection of the People's Republic of China. The Action Plan for Comprehensive Management of Air Pollution in BeiJing-Tianjin-Hebei and Surrounding Areas in Autumn and Winter 2017–2018. Available online: https://www.mee.gov.cn/gkml/hbb/bwj/201708/t20170831_420696.htm (accessed on 12 May 2021).
- 29. The State Council of the People's Republic of China. Three-Year Action Plan for Winning the Blue Sky Defense War 2018-2020. Available online: http://www.gov.cn/zhengce/content/2018-07/03/content_5303158.htm (accessed on 13 May 2021).
- Liang, J.; Langbein, L. Performance Management, High-Powered Incentives, and Environmental Policies in China. Int. Public Manag. J. 2015, 18, 346–385. [CrossRef]
- 31. Heimer, M. The Cadre Responsibility System and the Changing Needs of the Party. In *The Chinese Communist Party in Reform*; Brødsgaard, K.E., Zhang, Y.N., Eds.; Routledge: London, UK, 2006; pp. 122–138.
- 32. Chan, H.S. Cadre Personnel Management in China: The Nomenklatura System, 1990–1998. China Q. 2004, 179, 703–734. [CrossRef]

- Zhang, X. Implementation of Pollution Control Targets in China: Has a Centralized Enforcement Approach Worked? *China Q.* 2017, 231, 749–774. [CrossRef]
- 34. The State Council of the People's Republic of China. The Method for Assessing the Total Emissions Reduction of Major Pollutants. Available online: http://www.gov.cn/zwgk/2007-11/23/content_813617.htm (accessed on 13 May 2021).
- 35. Smith, G. Political Machinations in a Rural County. *China J.* **2009**, *62*, 29–59. [CrossRef]
- Wong, C.; Karplus, V.J. China's War on Air Pollution: Can Existing Governance Structures Support New Ambitions? *China Q.* 2017, 231, 662–684. [CrossRef]
- 37. Nahm, J. Exploiting the Implementation Gap: Policy Divergence and Industrial Upgrading in China's Wind and Solar Sectors. *China Q.* **2017**, *231*, 705–727. [CrossRef]
- 38. Eaton, S.; Kostka, G. Authoritarian Environmentalism Undermined? Local Leaders' Time Horizons and Environmental Policy Implementation in China. *China Q.* 2014, 218, 359–380. [CrossRef]
- Kostka, G. China's Evolving Green Planning System: Are Targets the Answer? Frankfurt School–Working Paper Series: Frankfurt, Germany, 2013; pp. 1–32. Available online: https://econpapers.repec.org/paper/zbwfsfmwp/201.htm (accessed on 20 June 2021).
- 40. Shen, W.; Jiang, D. Making Authoritarian Environmentalism Accountable? Understanding China's New Reforms on Environmental Governance. J. Environ. Dev. 2021, 30, 41–67. [CrossRef]
- 41. Zhang, P. Target Interactions and Target Aspiration Level Adaptation: How Do Government Leaders Tackle the "Environment-Economy" Nexus? *Public Adm. Rev.* 2021, *81*, 220–230. [CrossRef]
- 42. Hao, X.; Song, M.; Feng, Y.; Zhang, W. De-Capacity Policy Effect on China's Coal Industry. Energies 2019, 12, 2331. [CrossRef]
- 43. Concept Program Information Document (PID): Shanxi Energy Transition and Green Growth Development Policy Operation. Available online: https://documents.worldbank.org/en/publication/documents-reports/documentdetail (accessed on 30 April 2021).
- 44. The General Office of Shanxi Government. Shanxi Province's Action Plan for Comprehensive Management of Air Pollution in Autumn and Winter 2017–2018. Available online: http://www.shanxi.gov.cn/sxszfxxgk/sxsrmzfzcbm/sxszfbgt/flfg_7203 /bgtgfxwj_7206/201710/t20171012_338877.shtml (accessed on 7 June 2021).
- 45. The People's Government of Shanxi Province. Shanxi Province's Three-Year Action Plan for Winning the Blue Sky Defense War. Available online: http://www.shanxi.gov.cn/sxszfxxgk/sxsrmzfzcbm/sxszfbgt/flfg_7203/szfgfxwj_7205/201808/t20180806_ 468873.shtml (accessed on 7 June 2021).
- Chen, H.; Chen, W. Potential Impact of Shifting Coal to Gas and Electricity for Building Sectors in 28 Major Northern Cities of China. *Appl. Energy* 2019, 236, 1049–1061. [CrossRef]
- Meng, K.; Xu, X.; Cheng, X.; Xu, X.; Qu, X.; Zhu, W.; Ma, C.; Yang, Y.; Zhao, Y. Spatio-Temporal Variations in SO₂ and NO₂ Emissions Caused by Heating over the Beijing-Tianjin-Hebei Region Constrained by an Adaptive Nudging Method with OMI Data. *Sci. Total Environ.* 2018, 642, 543–552. [CrossRef]
- 48. Shin, K. Environmental Policy Innovations in China: A Critical Analysis from a Low-Carbon City. *Environ. Polit.* **2018**, *27*, 830–851. [CrossRef]
- 49. Xi, J. Full Text of Xi Jinping's Report at 19th CPC National Congress. Available online: https://www.chinadaily.com.cn/china/19 thcpcnationalcongress/2017-11/04/content_34115212.htm (accessed on 23 June 2021).
- 50. Kostka, G.; Zhang, C. Tightening the Grip: Environmental Governance under Xi Jinping. *Environ. Polit.* **2018**, 27, 769–781. [CrossRef]
- Xi's Speech Had 89 Mentions of the 'Environment,' Just 70 of the 'Economy'. Available online: https://www.bloomberg.com/ news/articles/2017-10-18/in-xi-s-vision-for-china-environment-edges-out-economy (accessed on 24 June 2021).
- 52. Li, K. Report on the Work of the Government (2017). 2017. Available online: http://english.www.gov.cn/premier/news/2017/0 3/16/content_281475597911192.htm (accessed on 3 July 2021).
- 53. Chen, G.C.; Lees, C. The New, Green, Urbanization in China: Between Authoritarian Environmentalism and Decentralization. *Chin. Polit. Sci. Rev.* 2018, *3*, 212–231. [CrossRef]
- 54. The Ministry of Environmental Protection of the People's Republic of China. Regulations on Quantitative Accountability for Action Plan Comprehensive Management of Air Pollution in Beijing-Tianjin-Hebei and Surrounding Areas in Autumn and Winter 2017–2018. Available online: https://www.mee.gov.cn/gkml/hbb/bwj/201708/t20170831_420711.htm (accessed on 4 June 2021).
- 55. The Ministry of Ecology and Environment Launched the "Thousand Mile Eye Plan" to Help Control Haze in Three Key Areas. Available online: http://www.china.org.cn/chinese/2018-08/27/content_60525605.htm (accessed on 2 August 2021).
- Miyamoto, A.; Ishiguro, C. *The Outlook for Natural Gas and LNG in China in the War against Air Pollution*; Oxford Institute for Energy Studies: Oxford, UK, 2018; pp. 1–53. Available online: https://www.oxfordenergy.org/publications/outlook-naturalgas-lng-china-war-air-pollution/ (accessed on 15 June 2021).
- 57. Natural Resource Defense Council (NRDC). China Dispersed Coal Governance Report 2018. Available online: http://nrdc.cn/ information/informationinfo?id=190&cook=1 (accessed on 17 June 2021).
- Government of Taiyuan City. The Key Measures for Comprehensive Management of Air Pollution in Taiyuan in Autumn and Winter 2017–2018. 2017. Available online: http://www.taiyuan.gov.cn/doc/2017/11/03/1182739.shtml (accessed on 5 June 2021).

- 59. Government of Yangquan City. Guidance on Clean Heating in Yangquan City in Winter 2017–2019. 2017. Available online: https://www.pkulaw.com/lar/557d61750d42f944cd5924f3d2e3f47bbdfb.html (accessed on 6 June 2021).
- 60. Government of Changzhi City. The Action Plan for Comprehensive Management of Air Pollution in Changzhi City in Autumn and Winter 2017–2018. 2017. Available online: https://www.changzhi.gov.cn/sjb/zwgk/zbwj_13339/201710/t20171031_415709 .shtml (accessed on 5 June 2021).
- 61. Government of Jincheng City. Implementation Plan of Winter Clean Heating Convernsion in Jincheng City. 2017. Available online: http://nyj.jcgov.gov.cn/sjbwz/hyfg/hyzc/202011/t20201122_1293784.html (accessed on 5 June 2021).
- 62. Government of Jinzhong City. Implementation Plan of "Replacing Coal with Gas" Project for Clean Heating in Jinzhong City in Winter 2017. 2017. Available online: https://www.sxjz.gov.cn/zwgk/fdzdgknr/zfwj/szfbgt43zfbgs/content_36427 (accessed on 6 June 2021).
- Government of Yuncheng City. The Action Plan for Comprehensive Management of Air Pollution in Yuncheng City in Autumn and Winter 2017–2018. 2017. Available online: https://www.yuncheng.gov.cn/doc/2019/10/30/24055.shtml (accessed on 5 June 2021).
- 64. Government of Linfen City. The Action Plan for Air Pollution Prevention and Control in Linfen City in 2017. 2017. Available online: http://www.linfen.gov.cn/gov/article_34490.html (accessed on 5 June 2021).
- Government of Lüliang. The Action Plan for Comprehensive Management of Air Pollution in Lüliang City in Autumn and Winter 2017–2018. 2017. Available online: http://www.lvliang.gov.cn/content/2017-10/19/content_221155.htm (accessed on 5 June 2021).
- Government of Datong. The Action Plan for Comprehensive Management of Air Pollution in Datong City in Autumn and Winter 2017–2018. 2017. Available online: http://www.dt.gov.cn/dtzww/bgtwj/201710/310b2d4dfad44f7f844af42ef52a3b42.shtml (accessed on 6 June 2021).
- 67. Government of Xinzhou. The Action Plan for Comprehensive Management of Air Pollution in Xinzhou City in Autumn and Winter 2017–2018. 2017. Available online: https://sthjj.sxxz.gov.cn/zlzt/dqwrzhzlqhdc/201710/t20171021_52317.html (accessed on 6 June 2021).
- Government of Shuozhou. The Action Plan for Comprehensive Management of Air Pollution in Shuozhou City in Autumn and Winter 2017–2018. 2017. Available online: http://www.shuozhou.gov.cn/xxgk/zbwj/201711/t20171108_160117.html (accessed on 6 June 2021).
- 69. The Ministry of Ecology and Environment of the People's Republic of China. The Action Plan for Comprehensive Management of Air Pollution in Beijing-Tianjin-Hebei and Surrounding Areas in Autumn and Winter 2018–2019. 2018. Available online: https://www.mee.gov.cn/gkml/sthjbgw/sthjbwj/201809/t20180927_630570.htm (accessed on 23 May 2021).
- 70. The Ministry of Ecology and Environment of the People's Republic of China. The Action Plan for Comprehensive Management of Air Pollution in Beijing-Tianjin-Hebei and Surrounding Areas in Autumn and Winter 2019–2020. 2019. Available online: http://www.mee.gov.cn/xxgk2018/xxgk/xxgk03/201910/t20191016_737803.html (accessed on 23 May 2021).
- 71. The Ministry of Ecology and Environment of the People's Republic of China. The Action Plan for Comprehensive Management of Air Pollution in Beijing-Tianjin-Hebei and Surrounding Areas, and Fen-Wei Plain in Autumn and Winter 2020–2021. 2020. Available online: http://www.mee.gov.cn/xxgk2018/xxgk/xxgk03/202011/t20201103_806152.html (accessed on 23 May 2021).
- 72. The Ministry of Ecology and Environment of the People's Republic of China. The Action Plan for Comprehensive Management of Air Pollution in Fen-Wei Plain in Autumn and Winter 2018–2019. 2018. Available online: http://www.mee.gov.cn/xxgk2018/ xxgk/xxgk03/201810/t20181029_667650.html (accessed on 23 May 2021).
- 73. The Ministry of Ecology and Environment of the People's Republic of China. The Action Plan for Comprehensive Management of Air Pollution in Fen-Wei Plain in Autumn and Winter 2019–2020. 2019. Available online: http://www.mee.gov.cn/xxgk2018/xxgk/xxgk03/201911/t20191112_741904.html (accessed on 23 May 2021).
- 74. Shanxi Province Clean Heating Work Leading Group. Investigation Report on Clean Heating in the Shanxi Province. 2021. Available online: http://nyj.shanxi.gov.cn/gzdt/tzgggs/202101/t20210126_1424337.html (accessed on 8 August 2021).
- 75. The Ministry of Finance of People's Republic of China. Notice on Carrying out Central Financial Support for the Pilot Work of Winter Clean Heating in Northern Areas. 2017. Available online: http://jjs.mof.gov.cn/zhengcefagui/201705/t20170519_260421 7.htm (accessed on 18 May 2021).
- 76. National Resource Defense Council (NRDC). China Dispersed Coal Management Report 2019. Available online: http://coalcap.nrdc.cn/datum/info?id=101&type=2 (accessed on 17 June 2021).
- 77. Department of Environmental Protection of Shanxi Province. Shanxi Environmental Statement 2017. 2018. Available online: https://sthjt.shanxi.gov.cn/hjzkgb/61340.jhtml (accessed on 17 May 2021).
- 78. Department of Ecology and Environment of Shanxi Province. Shanxi Province Ecology and Environmental Statement 2018. 2019. Available online: https://sthjt.shanxi.gov.cn/hjzkgb/81878.jhtml (accessed on 20 May 2021).
- 79. Department of Ecology and Environment of Shanxi Province. Shanxi Province Ecology and Environmental Statement 2019. 2020. Available online: https://sthjt.shanxi.gov.cn/hjzkgb/104476.jhtml (accessed on 12 May 2021).
- 80. Shanxi Province Clean Heating Work Leading Group Office. Notice on Early Completion of Clean Heating Conversion in 2021. 2021. Available online: http://nyj.shanxi.gov.cn/tzgg/83660.jhtml (accessed on 21 May 2021).
- 81. Shan, J.; Zhang, H. China Faces Gas Shortage in Winter Heating. Available online: https://www.globaltimes.cn/content/107980 9.shtml (accessed on 3 July 2021).

- 82. Zhou, V. Why China's Winter Fuel Crisis Is a Cold, Hard Lesson in the Law of Unintended Consequences. Available online: https://www.scmp.com/news/china/policies-politics/article/2124800/why-chinas-winter-fuel-crisis-cold-hard-lessonlaw (accessed on 3 July 2021).
- 83. Heating Supply Shortage: Heating Disrupted in Linfen, North China's Shanxi Province. Available online: https://news.cgtn. com/news/3d636a4e79594464776c6d636a4e6e62684a4856/share_p.html (accessed on 4 July 2021).
- Shanxi Province Energy Administration. Implementation of the Central Ecological Environmental Protection Inspector's "Looking Back" and the Special Inspector's Feedback Questions No. 31 for Air Pollution Prevention and Control. Available online: http://nyj.shanxi.gov.cn/tzgg/80527.jhtml (accessed on 18 July 2021).
- 85. Consultation List: Shanxi Province Development and Reform Commission. Available online: http://fgw.shanxi.gov.cn/zmhd/ zxts/zxlb/index_4.shtml (accessed on 30 June 2021).
- Liu, C.; Zhu, B.; Ni, J.; Wei, C. Residential Coal-Switch Policy in China: Development, Achievement, and Challenge. *Energy Policy* 2021, 151, 112165. [CrossRef]
- Qin, Y. Natural Gas in China's Power Sector: Challenges and the Road Ahead; Oxford Institute for Energy Studies: Oxford, UK, 2020; pp. 1–15. Available online: https://www.oxfordenergy.org/publications/natural-gas-in-chinas-power-sector-challenges-and-the-road-ahead/ (accessed on 27 July 2021).